Author's response to reviews

Title: An association study on contrasting cystic fibrosis endophenotypes recognizes KRT8 but not KRT18 as a modifier of cystic fibrosis disease severity and CFTR mediated residual chloride secretion

Authors:

Frauke Stanke (mekus.frauke@mh-hannover.de)
Silke Hedtfeld (hedtfeld.silke@mh-hannover.de)
Tim Becker (becker@imbie.meb.uni-bonn.de)
Burkhard Tümmler (tuemmler.burkhard@mh-hannover.de)

Version: 5 Date: 24 April 2011

Author's response to reviews:

Dear Sir,

as requested, all references are now placed in square brackets. The reference list is updated ensuring that all references are cited in consecutive order.

Yours sincerely,
Frauke Stanke